Author’s response to reviews

Title: Economic burden of symptomatic iron deficiency - A Survey among Swiss women

Authors:

Patricia Blank (patricia.blank@unibas.ch)
Yuki Tomonaga (yuki.tomonaga@uzh.ch)
Thomas Szucs (thomas.szucs@unibas.ch)
Matthias Schwenkglenks (matthias.schwenkglenks@uzh.ch)

Version: 1 Date: 19 Mar 2017

Author’s response to reviews:

Dear Editor

We have adapted the manuscript accordingly and added a paragraph and references regarding informed consent, ethical principles and the consumer panel in general (Methods/ Study design, line 87-94). We hope that this information is sufficient.

I can confirm that all authors, including myself, have all approved the final version.

We very much look forward to receiving your response.

Yours sincerely,

PD Dr Patricia R. Blank, MPH